Identifying features associated with higher-quality hospital care and shorter length of admission for people with dementia: a mixed-methods study

Rahil Sanatinia,^{1*} Mike J Crawford,^{1,2} Alan Quirk,² Chloe Hood,² Fabiana Gordon,³ Peter Crome,⁴ Sophie Staniszewska,⁵ Gemma Zafarani,¹ Sara Hammond,¹ Alistair Burns⁶ and Kate Seers⁵

¹Centre for Psychiatry, Imperial College London, London, UK

Declared competing interests of authors: Mike Crawford is Director of the College Centre for Quality Improvement at the Royal College of Psychiatrists (2011 to present) and was a member of the National Institute for Health Research Health Technology Assessment General Committee (2018-19). He also reports that Imperial College London has received other research grants from the National Institute for Health Research and other funding bodies. Alan Quirk and Chloe Hood work at the Royal College of Psychiatrists (London, UK) and oversee the National Audit of Dementia, since 2013 and 2008 respectively. Peter Crome is chairperson of the Steering Group for the National Audit of Dementia and has been on the Health Technology Assessment Primary Care, Community and Preventive Interventions Panel and Health Technology Assessment Prioritisation Committee A (Out of Hospital) (2014-19). He also reports that University College London has received other research grants from the National Institute for Health Research and other funding bodies. Sophie Staniszewska has been on the following committees: Health Services and Delivery Research Associate Board Members (2012-17); and Health Services and Delivery Research Researcher Led - Associate Board Members and INVOLVE Board (2005–12). She also reports that University of Warwick has received other research grants from the National Institute for Health Research and other funding bodies. Kate Seers was a member of the Health Services and Delivery Research Commissioning Board (Researcher Led) (2010-18). She also reports that University of Warwick has received other research grants from the National Institute for Health Research and other funding bodies.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published April 2020 DOI: 10.3310/hsdr08220

²College Centre for Quality Improvement, Royal College of Psychiatrists, London, UK

³Population Health Sciences, University of Bristol, Bristol, UK

⁴Research Department of Primary Care and Population Health, University College London, London, UK

⁵Warwick Research in Nursing, Warwick Medical School, University of Warwick, Coventry, UK

⁶Faculty of Medical and Human Sciences, University of Manchester, Manchester, UK

^{*}Corresponding author r.sanatinia@imperial.ac.uk

Plain English summary

Hospital care and stay for people with dementia Health Services and Delivery Research 2020; Vol. 8: No. 22

DOI: 10.3310/hsdr08220

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

oncerns have repeatedly been raised about the quality of care that people with dementia receive when they are admitted to hospital. A recent audit found that many people do not receive a proper assessment of their needs and that carers are not sufficiently involved during the person's admission or discharge. As a result, patients' needs may not be properly met and people can end up spending longer in hospital than they need to.

We set out to identify what hospitals can do to improve the quality of care that people with dementia receive and reduce the amount of time that people spend in hospital. We conducted a detailed analysis of information on > 10,000 patients and 4688 carers that was collected as part of a recent national audit. We also conducted a series of in-depth interviews with staff and carers of patients at six hospitals.

We found that people with dementia spend less time in hospital when discharge planning is started soon after their admission. To be effective, it is important that named staff take responsibility for co-ordinating the person's discharge and that systems are in place for getting support from more senior staff when problems arise. Clear communication between staff and family carers can help ensure that staff are aware of the patient's needs and that carers have realistic expectations about what the hospital staff can and cannot provide. When carers are present on the ward, they can reassure patients, encourage them to eat and drink and to accept the treatments they are offered. We also found that training on dementia needs to be provided to all staff in the hospital. If training is followed up by 'hands-on' support from senior staff, it can help to ensure that person-centred care is delivered in a timely manner.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 14/154/09. The contractual start date was in July 2016. The final report began editorial review in July 2019 and was accepted for publication in January 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Sanatinia *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of **Health Services and Delivery Research** and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk